Author's response to reviews

Title: Levels, Timing, and Etiology of Stillbirths in Sylhet District of Bangladesh

Authors:

Abdullah H Baqui (abaqui@jhsph.edu)
Yoonjoung Choi (yoonjoung@gmail.com)
Emma K Williams (emwillia@jhsph.edu)
Shams E Arifeen (shams@icddrb.org)
Ishtiaq Mannan (ismannan@savechildren.org)
Gary L Darmstadt (gary.darmstadt@gatesfoundation.org)
Robert E Black (rblack@jhsph.edu)

Version: 7 Date: 25 February 2011

Author's response to reviews:

February 25, 2011

BMC Pregnancy and Childbirth

Dear Managing Editor,

We submit a revised manuscript entitled, “Levels, timing, and etiology of stillbirths in Sylhet, Bangladesh”, for consideration for publication in the BMC Pregnancy and Childbirth (MS: 7481188193974381). We made all required format changes.

This paper has not been published and is not being considered for publication elsewhere. All authors have seen and approved the content and have contributed significantly to the work. The authors do not have a commercial or other association that might pose a conflict of interest. I will be the corresponding author to whom pre-publication matters should be addressed.

Thank you very much for your consideration.

Sincerely,

Abdullah H. Baqui, MBBS, MPH, DrPH
Professor
Department of International Health
Johns Hopkins Bloomberg School of Public Health
Email: abaqui@jhsph.edu
Tel: 410-615-3850